Author's response to reviews

Title: No evidence for association between polymorphisms in GRM3 and schizophrenia

Authors:

Nadine Norton (nortonn@cf.ac.uk)
Hywel J Williams (williamshj1@cf.ac.uk)
Sarah Dwyer (dwyersl@cf.ac.uk)
Dobril Ivanov (ivanovdk1@cf.ac.uk)
Anna C Preece (preeceac@Cardiff.ac.uk)
Amy Gerrish (gerrisha1@cf.ac.uk)
Nigel M Williams (williamsnm@cf.ac.uk)
Pamela Yerassimou (garethharcombe@hotmail.com)
Stanley Zammit (zammits@cf.ac.uk)
Michael C O'Donovan (odonovanmc@cf.ac.uk)
Michael J Owen (owenmj@cf.ac.uk)

Version: 6 Date: 11 May 2005

Author's response to reviews: see over
ref: MS: 1887460526146881. No evidence for association between polymorphisms in GRM3 and schizophrenia.

Dear Sir / Madam,

Thank you for accepting this manuscript for publication in BMC Psychiatry. The following minor essential and formatting revisions have been made as follows:

Reviewer 2.

1. “GRM3” has been italicized when meaning the gene throughout the manuscript. These changes were made on page 3, line 16, page 5, lines 10 and 13.

2. The closing bracket on the legend for figure one has been corrected.

Formatting changes requested.

3. The full stop has been removed from the title on both the manuscript and in the submission system.

4. The author contributions have been checked and are correct.

5. The table legends have been moved and now follow each table.

6. The web address for the international hapmap project has been removed from the manuscript text and is cited in the references.